Author's response to reviews

Title: The RISAP-study: a complex intervention in risk communication and shared decision-making in general practice

Authors:

Pia Kirkegaard (pia.kirkegaard@alm.au.dk)
Adrian GK Edwards (edwardsag@cf.ac.uk)
Bo Hansen (bo.hansen@alm.au.dk)
Mette D Hansen (mdh@alm.au.dk)
Morten SA Jensen (msa.jensen@gmail.com)
Torsten Lauritzen (tl@alm.au.dk)
Mette B Risoer (mettriso@rm.dk)
Janus L Thomsen (jlt@dadlnet.dk)

Version: 4 Date: 20 September 2010

Author's response to reviews:

Abstract and email addresses have been added. One email address has been changed: Mette Risoer: mette.bech@uit.no (before: mettriso@rm.dk)